



Contacts at the Nanoscale and for Nanomaterials

Hei Wong ^{1,*}, Jieqiong Zhang ² and Jun Liu ²

- ¹ Department of Electrical Engineering, City University of Hong Kong, Hong Kong, China
- ² Hubei Jiu Feng Shan Laboratory, Wuhan 430074, China; zhangjieqiong@jfslab.com.cn (J.Z.);
 - liujun@jfslab.com.cn (J.L.)
- * Correspondence: heiwong@ieee.org

Abstract: Contact scaling is a major challenge in nano complementary metal–oxide–semiconductor (CMOS) technology, as the surface roughness, contact size, film thicknesses, and undoped substrate become more problematic as the technology shrinks to the nanometer range. These factors increase the contact resistance and the nonlinearity of the current–voltage characteristics, which could limit the benefits of the further downsizing of CMOS devices. This review discusses issues related to the contact size reduction of nano CMOS technology and the validity of the Schottky junction model at the nanoscale. The difficulties, such as the limited doping level and choices of metal for band alignment, Fermi-level pinning, and van der Waals gap, in achieving transparent ohmic contacts with emerging two-dimensional materials are also examined. Finally, various methods for improving ohmic contacts, junction doping technology, phase and bandgap modification effects, buffer layers, are highlighted.

Keywords: metal/semiconductor contact; Schottky contact; ohmic contact; 2D material/metal contacts; contact scaling

1. Introduction

Contacts, referring to a metal's connection to a semiconductor or insulating film, are integral parts of all electronic devices. To exploit the electronic properties of electronic devices to a large degree, it is essential to produce ohmic contacts that allow for efficient and nondistorted signal transmission. Over decades of significant effort, solid theoretical accomplishments related to metal/semiconductor interfaces, and Schottky junction charge transport studies, as well as developments in the technological know-how for achieving good ohmic contacts, during the period from the 1930s to 1950s, contacts for microelectronic devices were not considered a major challenging issue. In the last couple of decades, contacts reappeared as a challenging issue with the aggressive downsizing of CMOS devices and the introduction of 2D materials, such as graphene, carbon nanotubes, and 2D transition metal dichalcogenides.

In the era of metal–oxide–semiconductor (MOS) or complementary MOS (CMOS) technology, many efforts have been devoted to the technology and the structural advancement of devices for the continual downsizing of the gate length and increasing of the integration density [1–5]. As the feature size of CMOS technology approaches a few nanometers, which is considered to be the physical and technological limits, shortening of the gate length has faced ever-tougher challenges and the pace of downsizing has slowed down in recent years (see Figure 1). The scaling rule and technology node assignment have shifted from physical gate length to equivalent gate length [6–8]. For instance, for the 130 nm technology node, the physical gate length is 65 nm, and the half-pitch width is 150 nm, implying that the widths of the spacer and the source–drain regions are approximately 85 nm. For technology beyond the 45 nm technology node, the gate length is 38 nm, and the half-pitch width is 68 nm [5], indicating a more aggressive scaling of the

Citation: Wong, H.; Zhang, J.; Liu, J. Contacts at the Nanoscale and for Nanomaterials. *Nanomaterials* **2024**, *14*, 386. https://doi.org/10.3390/ nano14040386

Academic Editor: Mircea Dragoman

Received: 23 January 2024 Revised: 6 February 2024 Accepted: date Published: 19 February 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/licenses/by/4.0/). contact and spacer regions in later technology nodes. The source–drain contact regions have been narrowed to reduce the half-pitch size of the MOS transistor even when the gate length is not scaled according to Moore's Law. With the FinFET technology introduced for the 28 nm technology node, the technology node assignment was based on the density level rather than the physical gate length. The 3D structure of FinFET and the larger effective gate width with a small footprint enables a higher chip density [8–10]. More nonclassical or non-Denard strategies [4], such as GAA/nanosheets, CFET, contact size reduction, cell size reduction, back power rail, buried power rail, back interconnection, nano TSV, and stacking or heterogeneous 3D packaging [11] (see Figure 1), will eventually be adopted to extend Moore's Law to the subnanometer range.



Figure 1. Illustration of more Moore strategies for downsizing effective technology nodes beyond the decananometer and subnanometer ranges.

Figure 2 shows the trend in contact scaling toward 1 nm technology. If we assume the spacer width is 8 nm, then one can estimate that the contact width for 5 nm technology is approximately 22 nm, which is the most substantial area of the whole transistor. According to the technological roadmap, as expected from the ITRS, the spacer width will be 6 to 5 nm in the coming technology nodes. An aggressive size reduction of the contact area is expected. The source and drain contact areas of FinFET structures have decreased from more than 1300 nm² to approximately 500 cm², respectively, in the downsizing of the 20 nm technology node to the 5 nm technology node [12]. However, to maintain the performance gains achieved with FinFET technology below the 40 nm CGP range, the contact resistivity should be smaller than $10^{-9} \Omega$ -cm². This is a challenging issue. A metal/semiconductor contact based on thermionic emission does not seem to be able to provide such a low resistivity value. In addition, it was predicted that for 30 nm CGP technology, the gate stack's metal film thickness also needs to be scaled down to 6 nm [13], and the source/drain contacts for the nanosheet channel should be limited to the several decananometer range. The surface roughness of the thin films and their contact with silicon will have some adverse effects. We provide an in-depth discussion of these issues in Section 3.



Figure 2. (a) Definition of contacted gate pitch (CGP) or contact poly pitch (CPP); (b) trends in gate length, contacted gate length, and contact area downsizing. Data taken from [5,12].

On the other hand, two-dimensional materials are considered promising silicon replacements when silicon-based CMOS devices reach their physical limits [14-20]. A recent focus related to 2D materials are transition metal dichalcogenides (TMDs), such as molybdenum disulfide (MoS2) [21-23], ditelluride (MoTe2) [24,25], tungsten diselenide (WSe2) [26,27], and phosphorene. These emerging 2D materials have an acceptable gap value, which is unavailable in the first 2D material discovered: graphene. Like graphene, each monolayer of these materials is connected via the van der Waals force. These materials exhibit a number of attractive properties, such as high electron mobility, a low optical absorption coefficient, and high electrical and thermal conductivities. These advanced features could enhance the device and circuit performance in various aspects. However, there are still unresolved challenges related to mass production, complex device design, and circuit integration [20]. One of the challenges is achieving good ohmic contacts between 2D materials and metals. Unlike silicon, 2D materials have no dangling bonds and can form ideal contacts with metals. However, because only a few monolayers are thick and the dopant concentration is far less than that of conventional semiconductors, the contact resistances of metal/2D materials are much larger than those of conventional semiconductors. In addition, the new materials have different bandgap and band offsets, which require nonconventional metals to achieve the required energy level alignment for ohmic contact. In Section 4, we explore the physics of metal/2D material contacts, recent progress, and process options for tackling ohmic contact issues.

This review aims to provide a comprehensive overview of metal/semiconductor Schottky contact theory and its applications to 2D materials. Section 2 introduces the basic concepts, theoretical background, and assumptions used to develop the current–voltage model for Schottky contacts. We address some common misconceptions or inappropriate applications of the Schottky equation to the metal–insulator–semiconductor (MIS) structures. Section 3 examines the criteria and methods for achieving ohmic contacts and the scaling effects for contacts down to the decananometer range. Section 4 summarizes the recent advances in developing ohmic contacts for 2D materials, focusing on comparing metal van der Waals and hybridization contacts. We also review the techniques for junction doping, bandgap engineering, and phase engineering of 2D materials for contact optimization. Furthermore, we discuss the role of buffer layers in mitigating the Fermi-level pinning effects and enhancing the contact characteristics. Section 5 explores the characteristics of 2D material/Si contacts. Two-dimensional materials on silicon have been recognized as a feasible prior technology for the widespread application of 2D materials, which offers a promising way to leverage the advantages of both materials.

2. Schottky Junction

Long before the invention of the transistor, rectifying contacts were discovered and became a research hot spot [28]. It is known that the contact between a metal and a semiconductor has different resistance values depending on the polarity. This property led to the development of crystal rectifiers, such as copper–cuprous oxide rectifiers, which were widely used in early AM radio receivers for signal detection. Various theories have been proposed to model the rectifying characteristics. In particular, works by Nordheim [29], Frenkel [30], Fowler [31], Schottky [32], and Mott [33] have established a solid foundation for modern semiconductor physics, especially for current conduction across metal–semiconductor and metal–tunnel-oxide–semiconductor contacts. Schottky and Mott state that the first principle of band alignment and, thus, the barrier height is when two different materials are brought into contact. The rectifying contact resulting from a metal/semiconductor connection was later named a Schottky junction for the significant contribution of this widely accepted model.

2.1. Schottky Equation

The key component of the Schottky equation is the thermal emission law, as proposed by Richardson [34], which was based on experimental results related to the current density generated from a heated metal wire. The temperature and current dependencies are as follows:

$$J = A^* T^2 \exp\left(\frac{W}{kT}\right) \tag{1}$$

where *W* is the workfunction of the metal, kT is the thermal voltage, and A^* is the effective Richardson constant. Schottky later modified the expression to include the effect of an electric field, and its final form is now usually expressed in terms of an applied voltage, *V*, as follows [32,33]:

$$V_{Schottky} = A^* T^2 \exp\left(-\frac{\Phi_{\rm SB}}{kT}\right) \exp\left(\frac{qV}{kT}\right)$$
(2)

where *q* is the electron charge, and Φ_{SB} is the Schottky barrier.

In the original proposal by Schottky and Mott [32,33], the Schottky barrier is as follows"

$$\Phi_{SB} = \Phi_M - \chi_S \tag{3}$$

Equation (3) provides the energy difference between the metal workfunction, Φ_M , and the semiconductor affinity, χ_S . The relationship is known as the Schottky–Mott rule, which describes the alignment of the energy bands when the two different materials have isolated state energies. For the case depicted in Figure 3a, an n-type semiconductor with an electron affinity smaller than the metal workfunction, equilibrium is achieved by transferring the electron on the semiconductor's surface to the metal, as the metal workfunction is larger than the electron affinity of the semiconductor. The amount of band bending on the semiconductor's surface is the same as the energy difference. However, the Schottky–Mott rule is not always accurate in predicting the barrier height. The Fermi level of the system tends to be pinned to a fixed position within the semiconductor bandgap, regardless of the metal used. This is because the charge distribution at the metal/semiconductor interface is not simply the sum of the charges on the isolated surfaces. There is also physical bonding or orbital overlap between the metal and semiconductor atoms, which changes the surface energy levels from their original values. This phenomenon is called



Fermi-level pinning [35–37]. Equation (4) approximates the Schottky barrier height with the FLP effect.

Figure 3. Metal/n-type semiconductor Schottky contact: (a) equilibrium; (b) forward bias; (c) reverse bias.

$$\Phi_{SB} = (S \times \Phi_M - \chi_S) + (1 - S)\Phi_{\text{MIGS}}$$
(4)

where Φ_{MIGS} is the metal-induced gap states (MIGS). The intrinsic surface states and interface dipole of the semiconductor should lead to the same consequence. The parameter S is the pinning factor, which reflects the strength of the pinning and can be determined by the slope of the barrier height versus metal workfunction plot, expressed as follows:

$$S = d\Phi_{SB}/d\Phi_M \tag{5}$$

The barrier height does not depend on the metal workfunction when S = 0. This is the Schottky–Mott limit when S = 1. However, S is usually much smaller than one for most common semiconductors. Figure 4 shows how the barrier height changes with different metal workfunctions from various sources [36,37]. The slope deviates significantly from the ideal Schottky–Mott rule.



Figure 4. Plot of the experimental Schottky barrier heights for different metals on n-type Si taken from various sources. The straight line indicates the Schottky–Mott rule [36]. © 1993 American Vacuum Society. Reproduced with permission.

The effective Richardson constant was later treated as a universal constant or the fundamental Richardson constant, *A*₀, which can be calculated by finding the ensemble of electrons leaving the metal surface at a kinetic energy exceeding the Fermi level of the metal, where the energy distribution of the electrons in the metal follows the Fermi–Dirac distribution. The fundamental Richardson constant is constituted by fundamental parameters, as follows:

$$A_0 = \frac{4\pi q m^* k^2}{h^3}$$
(6)

where m^* is the effective mass of an electron, and *h* is Planck's constant.

The fundamental Richardson constant, A₀, equals 120 A/cm²/K² when the effective mass is equal to the fundamental value of the electron mass. However, the effective Richardson constant, A^* , extracted from the experimental results using (2), always differs from the fundamental value, A₀. In the Schottky diode, the thermionic emission from the semiconductor, the Richardson constant, is further modified to account for the effective mass of different bands and band structures of various materials [38]. However, experimental results over the last 120 years have not shown that A^* is a universal constant, even when taking the effective mass into consideration for the case of thermionic emissions from semiconductors. Instead, it is often treated as an empirical parameter or a fiction of A₀ in many cases. Yet it is sound to consider the constant of a material-dependent parameter because of the different carrier velocities or lifetimes resulting from different or same materials prepared under varying conditions. In addition, the emitted electron may be reflected back to the electrode if we consider the carrier transport as wave propagation. Yet the aggressive scaling of the junction size in terms of the cross-section and thickness and the introduction of 2D material could offer an excellent opportunity to further disclose the mystery of the effective Richardson constant.

The second issue in applying the Schottky equation is that there is a vast list of publications, including numerous reports on 2D materials based on Schottky diodes or MIS Schottky diodes, that characterize the Schottky current with the "ideality factor", *n*. Taking the ideality factor into consideration, the Schottky equation becomes [39], as follows:

$$T = A^* T^2 \exp\left(-\frac{\Phi_{SB}}{kT}\right) \exp\left(\frac{qV}{nkT}\right)$$
(7)

The ideality factor was first defined in a pn junction diode, which has a solid physical ground, whereby if the forward current is contributed by the diffusion current only, n = 1, and when carrier generation and recombination (GR) take place, n = 2 [39]; n can increase

up to 4 if the generation–recombination involves multiple energy levels of defects [40,41]. Thus, n = 1 represents an ideal diode, and n > 1 indicates the contribution of a GR current. The ideality factor indicates the quality of the pn junction diode. In a Schottky diode, although the introduction of n can help to improve the fitting of the voltage dependence in some cases, it is not an indicator of junction quality. It does not have a sound physical or technical meaning in many cases. The n factor in the Schottky diode equation should be a reflection of the effectiveness of the barrier surface potential varying with the applied voltage. We further elaborate on this issue in Section 2.2.

2.2. Current Conduction in an MIS Diode

The Schottky equation can be used to model the current conduction of an MIS diode. However, this equation is based on some assumptions that are often overlooked. It is also suggested that, in some early metal/semiconductor contacts, there may exist some native thin oxide on the semiconductor's surface. Fowler suggested that there was a layer of "bad semiconductor" between a metal and a good semiconductor [31]. Although these proposals were later refuted, it did point to the importance of the interface's quality in the contact behavior. A similar situation arose in the recent study of 2D material contacts and 2D material/Si contacts, in which a thin tunneling layer may be present between the metal and the 2D material. In this case, the Schottky equation can still be used to approximate the tunneling current in the forward region but with a modified effective Richardson constant that accounts for the tunneling barrier [42]. The modified Schottky equation is as follows:

$$J_{MIS} = A_0 T^2 \exp(-\sqrt{\chi}\delta) \exp\left(-\frac{\Phi_{SB}}{kT}\right) \exp\left(\frac{qV}{nkT}\right)$$
(8)

where is the χ mean barrier height value between the insulator and the semiconductor and δ is the tunneling oxide thickness. We can define a new effective Richardson constant as follows:

$$A^* = A_0 \exp(-\sqrt{\chi}\delta) \tag{9}$$

Note that the current conduction is, in fact, dominated by the direct tunneling of the carriers over the thin insulating layer. The approximation of the Schottky equation involves modifying the Richardson coefficient with a transmission coefficient, which depends on the barrier height between the silicon and the insulator and the insulator thickness. These two factors are the key parameters that govern the conduction current. In addition, some effects, such as the minority carrier injection, image force, and quantum mechanism reflection, are also neglected. This approximation is only valid for a forward current with V > 3kT/q, because the metal/insulator barrier differs from the silicon/insulator barrier. Some authors have ignored these assumptions and treated the MIS diode as a regular Schottky diode.

Instead of treating the *n* factor as an "ideality factor" in (8), it was proposed that the nonunity *n* factor could be calculated from the surface potential of the barrier [42]. Defining *n* as a change in the surface potential, ΔV_s , with respect to the applied voltage (i.e., $n = -V/\Delta V_s$), and further taking the interface states into consideration, the electrostatic calculation leads to the following:

$$n = 1 + \frac{\left(\frac{\delta}{\varepsilon_i}\right)\left(\frac{\varepsilon_s}{W} + qD_s\right)}{1 + qD_m\delta/\varepsilon_i} \tag{10}$$

where α and α are the permittivities of the insulator and semiconductor, respectively; *W* is the depletion layer in the semiconductor surface; and D_s and D_m are the surface state densities in the equilibrium with the semiconductor and metal, respectively. If the surface state is negligible, (10) reduces to the following:

$$n = 1 + \frac{\delta/\varepsilon_i}{W/\varepsilon_s} \tag{11}$$

Thus, large n values, in some cases, are not mysterious; it is only because the electrostatic thickness of the insulating layer, δ/ε_i , is much larger than the electrostatic thickness of the depletion layer width, W/ε_s . If we neglect the surface states on the metal side, which is the case for most MIS diodes, Equation (10) reduces to the following:

$$n = 1 + \frac{\delta}{\varepsilon_i} \left(\frac{\varepsilon_s}{W} + q D_s \right) \tag{12}$$

In this expression, n is a parameter that describes how effectively the barrier surface potential varies with the applied voltage. It has nothing to do with the generation and recombination of carriers or the validity of the Schottky current model. Therefore, it is not an indicator of the ideality of the junction.

In fact, the validity of (8) for the MIS diode is limited. The conduction current should be more precisely modeled by the direct tunneling current (see Equation (13)) if the insulator layer and barrier height are in the direct tunneling range [43,44].

$$J_{DT} = J_0 \left(1 - \frac{V}{2\Phi_{\rm B}} \right) exp \left\{ -\frac{4}{3} \frac{\sqrt{2m_i q \delta} \Phi_{\rm B}}{\hbar V} \left[1 - \left(1 - \frac{V}{\Phi_{\rm B}} \right)^{3/2} \right] \right\}$$
(13)

where m_i is the electron mass in the oxide; \hbar is the reduced Planck's constant; and Φ_B is the barrier height (in electron volts) between the emitting electrode and the oxide.

For the case of thicker insulators or with a smaller voltage applied, tunneling over the triangular edge (see Figure 3c) is also possible. In this situation, the current conduction is better described with the Fowler–Nordheim (FN) equation, as shown below [45]:

$$J_{FN} = AE^2 \exp\left(-B/E\right) \tag{14}$$

$$A = \frac{q^3}{8\pi h \Phi_B} \left(\frac{m_e}{m_i}\right) \tag{15}$$

$$B = \frac{4}{3} \frac{(2m_i)^{1/2}}{q\hbar} \Phi_{\rm B}^{3/2} \tag{16}$$

where m_e is the electron mass in the free space.

The FN formula has been widely used to explain the conduction behavior of various thin dielectric films [45]. However, in contact studies including the so-called MIS Schottky diode, the current–voltage characteristics were often fitted with the modified Schottky Equation (8) or even the simple Schottky equation in (7) instead of the FN relationship. It has to be pointed out that the barrier height extracted from these fitting should be inaccurate, and the ideality factor does not carry any technical implications. The effective Richardson coefficient is also inappropriate in these cases. The current–voltage characteristics are better described with the FN equation.

3. Ohmic Contact in the Nanoscale

Ohmic contact refers to a metal-semiconductor contact with low and constant resistance regardless of the applied voltage polarity. A low contact resistance and nonrectifying contact enable signals to transmit into and out of a semiconductor device, such as transistors, LEDs, and solar cells, with minimum distortion. Figure 5a,b illustrate the idea for achieving an n-type ohmic contact. A small contact barrier can be obtained by choosing a small metal workfunction. Under forward bias, the electron can be emitted through the barrier via thermionic emission or Schottky emission. With reverse bias (see Figure 5b), the barrier height is increased, but a large reverse current is still possible if the barrier width is narrow enough so that direct tunneling is possible. The ohmic contact has not been considered a server-challenging issue since the start of silicon technology. The key factor governed by the contact resistance can be readily solved with the well-developed metallization and junction doping technique. The narrow barrier is achieved by heavily doping the contact region of the semiconductor. In addition, the junction dimensions, including the cross-sectional area, junction depth, and metal thickness, are quite large compared to the devices themselves. Figure 5c illustrates the ideal ohmic contact's characteristics, practical ohmic contact, and the scaling effect of the contact. As mentioned, to reduce the signal loss, the contact resistance should be as small as possible. The currentvoltage (I-V) relationship in the prescribed voltage range should be linear so that it does not cause signal distribution and harmonics. However, as illustrated in Figure 5a,b, barriers exist in the metal/semiconductor contact; the I-V characteristics are either governed by thermionic emission or tunneling, which are nonlinear. In addition, because the forward and reverse conductions are controlled by different mechanisms, the I-V characteristics are asymmetric. The degrees of nonlinearity and asymmetry may not cause significant issues when the signal levels are high. However, they could significantly downgrade the device's performance when the signal level is low. The contact's characteristics will further seriously deteriorate if the contact size is reduced, the signal level is lowered, and, in some cases, new materials and fabrication processes are introduced in nanoscale CMOS technology.



Figure 5. Ohmic contact and current-conducting mechanisms: (**a**) forward bias; (**b**) reverse bias. (**c**) Illustration of the current–voltage characteristics of an ideal ohmic contact, practical ohmic contact, and poor ohmic contact that may result from downsizing.

Figure 6a shows the key parameters and process issues for producing a good ohmic contact. Generally, ohmic contact has not been a major challenge in CMOS technology over the past six decades. The contact size is still quite large compared to the conductive current and applied voltage. Moreover, a suitable metal workfunction is available, and the semiconductor can be doped to the degeneracy level. However, in the nano CMOS technology, the situation is changed. To reduce the chip size, every dimension has been scaled

down. In the "2 nm" technology, even the power rails of the integrated circuit are scaled and moved to the back side of the IC. Reducing the contact's cross-sectional area is undoubtedly also an attractive option for fitting more transistors in the chip. Some new technology options, such as the use of high-k gate dielectric materials, will limit the choice of materials for metal gates [6]. Introducing a nanosheet device structure will result in an ultra-shallow junction, and heavy junction doping may not be feasible. The thickness of the metal layer also needs to be reduced. Consequently, the contact interface and the surface roughness of the metal film can become significant factors affecting the contact quality. The process consequences of contact scaling for nano CMOS technology are illustrated in Figure 6b.





Figure 7 shows how the physics of a Schottky junction and an ohmic contact can be affected by scaling down the junction size. The main effects are as follows:



Figure 7. (a) Possible effects resulting from a scaled metal/semiconductor contact; (b) issues associated with a contact with an interlayer.

- 1. Barrier lowering: The barrier height at the interface can be reduced by various factors, such as image force effects, metal workfunction variation, and surface roughness. These factors may be negligible in a large junction, but they can have significant impacts in a scaled junction. They can increase the forward current by lowering the potential barrier.
- 2. Barrier widening: The use of heavily doped contacts may not be possible in a scaled junction. This can result in a wider tunneling barrier, which reduces the conduction current under reverse bias.
- 3. Richardson constant reduction: Thinner metal films are used in a scaled junction, which can lead to a smaller value of the Richardson constant. This can decrease the forward current by reducing the thermionic emission.
- 4. Metal workfunction lowering: experimental results suggest that thinner metal films have lower workfunctions [46], which can also reduce the barrier height at the interface.
- 5. Interface states: The presence of interface states can cause Fermi-level pinning, which affects the barrier height and the band bending. This effect is more pronounced in thinner films and in unpassivated surfaces.
- 6. Interface layer: To mitigate the Fermi-level pinning effect or to enable different circuit design options, such as using 2D materials as interlayer conductors (see Section 4.5), an interface layer may be used for passivation. This can affect the reverse currents by changing the tunneling characteristics. The thickness, band offset, and dielectric constant of the interface layer are important parameters for this effect.

3.1. Effects of Junction Doping

The unavailability of a heavily doped contact results in a wider barrier. As shown in Figure 7a, this reduces the tunneling efficiency. Following the method developed by Yu [47], we can calculate the contact resistance depending on the dopant concentration. Figure 8 shows the contact resistance as a function of the dopant concentration. For high dopant concentrations (> 2.5×10^{19} cm⁻³), the contact resistance decreases to the range of 10^{-5} to $10^{-6} \Omega$ -cm². The current conduction is mainly due to field emission. For low-level doping (< 10^{18} cm⁻³), the contact resistance increases by 5 to 6 orders of magnitude. The current conduction is mainly governed by thermionic and field emissions. Hence, one should be aware that the Schottky equation is not always valid, especially when the doping concentration is high. The field emission needs to be taken into account. On the other hand, the contact resistance can be significantly reduced when field emission takes place. Thus, in a nanoscale contact for a 30 nm CPP [12], the ohmic contact regime should shift from the thermionic region to the field emission and tunneling regimes.



Figure 8. Illustration of doping concentration dependence of contact resistance involving different current conduction mechanisms.

3.2. Effects of Interface Roughness

The Schottky current of a metal/semiconductor can exceed the Schottky–Mott limit. Fundamental calculations show that the metal surface flatness affects the electron cloud near it [48,49]. It was found that the ratio of electric field fluctuations to the average electric field, denoted by $\delta E/E_s$, can be estimated by the following [49]:

$$\frac{\delta E}{E_S} = \Delta^2 \int_0^{k_c} \frac{(l_{cor}k)^2}{(1+a(l_{cor}k))^{1+r}} dk$$

where E_s denotes the electric field resulting from a smooth surface; δE is the increased electric field due to the surface roughness; k is related to the surface wave vector; Δ is the normalized roughness; l_{cor} is the normalized correlation length; r is the roughness exponent, which is a measure of the degree of surface irregularity; and a is a proportional constant. See Figure 9 for the definitions of the parameters.



Figure 9. Definitions of the key roughness parameters and an illustration of the local averaged thickness variation for small-sized devices as a function of the applied field [50].

Figure 9 shows how the roughness parameter, Δ , and the correlation length, *l*_{cor}, are normalized by the film thickness, *t*_{diel}. The normalized roughness Δ (= r_s/tdiel) becomes

more important for films with similar thicknesses. The correlation length, which measures the local field variation, is also inversely related to the film thickness, i.e., $l_{cor} = L_{cor} / t_{diel}$. This means that thinner films have larger values of Δ and l_{cor} and, therefore, larger electric field fluctuations [50]. However, the surface roughness is not a scalable parameter. When the film thickness reaches the atomic scale, the cluster size of a polycrystalline structure affects the surface roughness negatively. The increased local field lowers the effective barrier of the contact, which results in a significant rise in the Schottky emission, Fowler–Nordheim (FN) tunneling, and Poole–Frenkel (PF) emission [49,50].

One of the factors that affect the electrical properties of metal–insulator–metal (MIM) structures is the surface roughness of the interfaces. The lower interface (insulator on metal) tends to be rougher than the upper interface (metal on insulator) due to the crystalline nature and the grain size of the metal films, as well as the limitations of the deposition methods, such as evaporation or sputtering. This phenomenon has been known for decades and persists even in MIM structures with larger dimensions and thicker films [51–56]. Figure 10 shows an example of how the surface roughness influences the current conduction of MIM structures. The experiment was performed on TiN/Al₂O₃/TiN MIM capacitors with identical top and bottom electrodes. Ideally, the Schottky barriers should be equal, resulting in a symmetric current–voltage characteristic. However, Figure 10b reveals that the measured I-V characteristic is asymmetric. This is because the bottom and top barriers have different values: 3.01 eV and 3.65 eV, respectively (see Figure 10a).



Figure 10. (a) Comparison of the surface thickness fluctuations of a nude TiN sample, TiN with a 20 nm thick Al₂O₃ deposited by 150 °C ALD, and TiN with a 20 nm thick Al₂O₃ deposited by 330 °C ALD; (b) typical current–voltage characteristics of TiN/Al₂O₃/TiN MIM capacitors showing the asymmetric forward and reverse leakage currents and bottom electrode dependencies; (c) extraction of the bottom barrier height from the temperature-dependent forward I-V characteristics; (d) top barrier from the reverse characteristics [52]. © 2018 Elsevier. Reproduced with permission.

The electrical properties of metal/semiconductor contacts strongly depend on the metal film thickness and, also, the method of deposition [46]. These effects may be partly

related to the surface roughness. Figure 11 shows how the Schottky barrier height and Richardson constant vary with the metal thickness and deposition process. For an evaporated sample with a 100 Å thickness (see Figure 11a), the Cu electrode has a barrier height of approximately 0.58 eV and a Richardson constant close to the theoretical value of 112 A/cm²/K². Both parameters increase sharply and reach their saturation values of about twice the theoretical value for the Richardson constant and 0.62 eV for the barrier height for films thicker than 200 Å. For the sputtered Cu film (see Figure 11b), the theoretical Richardson value is maintained up to 200 Å. It increases almost exponentially as the film becomes thicker. The Richardson constant rises to about 1700 A/cm²/K², which is approximately 15 times the theoretical value, and the barrier height to 0.71 eV for 800 Å thick Cu films. Toyama attributed the difference to the high kinetic energies of Cu atoms during sputtering but did not provide further explanation. It is unclear how the kinetic energies of metal atoms can affect the Schottky characteristics. Unlike thermal evaporation, sputtering involves metal atoms with much higher kinetic energy. However, these results should be caused by other factors. Generally, the sputtering process produces a smaller grain size of the deposited metal clusters than the evaporated ones. Another important factor is that thin film sputtering is usually conducted in a low vacuum, resulting in a higher oxygen content in the film than evaporation. Evaporated films have larger grain sizes than sputtered films, which leads to a higher surface roughness. This could account for the lower barrier height observed in the evaporated films than the sputtered ones. The same mechanism can also explain the effect of the film thickness. A thicker film would have a smoother surface and, thus, a higher barrier height.



Figure 11. The metal thickness and deposition process are dependent on the Schottky barrier height and Richardson constant: (**a**) Cu-Si contacts by evaporation; (**b**) sputtered Cu-Si contacts prepared by sputtering. Redrawn based on [46].

Furthermore, because of the high oxygen content in the sputtered films, the electrons' film will be strongly bounded to the oxygen because of the greater electronegativity of the oxygen; as a result, it makes the thermionic emission of electrons more difficult. Theoretical calculations have shown that the workfunction of the Cu-O system is larger than that of pure copper [57]. The experimental results show that both the barrier height and the effective Richardson constant decreased. Factors such as the barrier height inhomogeneity and effective mass variation may have some effects [36,37], but they cannot, however, account for such a large difference in the effective Richardson constant. Because the effective Richardson constant has the same trend as the barrier height in terms of the thickness dependencies, one may infer that the effective Richardson constant is a function of the barrier height. Nevertheless, these results imply that the conduction current of an ultrathin metal/semiconductor contact at low voltage is lower than that of a thick metal contact. However, at high voltage, the situation is reversed, and the thin contact has a higher conduction current than the thick contact. Therefore, the nonlinearity of the I-V characteristics will be further deteriorated. The process dependence also becomes more significant as smaller size contacts and thinner metals are used. In the nano CMOS process, the metallization will gradually be replaced by the atomic layer deposition (ALD). The ALD process can produce films with high thickness uniformity and conformal coverage. However, ALD metal films are usually polycrystalline and have large grain sizes. This implies that the surface roughness will be high if the film thickness and contact cross-section are in the order of several hundred nanometers [58]. Figure 12 illustrates some examples of copper films deposited by the ALD process; in the worst case, the RMS roughness is 21 nm for a 4.7 nm thick film. This high roughness of the metal film would result in significant barrier lowering and lead to a large fluctuation in the contact resistance. Thus, preparing a metal film with low roughness could be the key issue for smaller-sized contacts.



Figure 12. AFM picture of atomic layer deposition of copper on Ru substrates: (**a**) 5.1 nm with an RMS roughness of 1.5 nm; (**b**) 3.4 nm thick with an RMS roughness of 3.4 nm; (**c**) 4.7 nm thick with an RMS roughness of 21 nm [58]. © 2016 Electrochemical Society. Reproduced with permission.

In summary, contacts have become one of the major challenges in nano CMOS technology. The challenge comes from the requirement of achieving an even lower contact resistance with the aggressively reduced contact area and film thickness. This challenge is exacerbated by the device structure's evolution, such as the transition from planar to Fin-FET and nanosheet architectures. Moreover, the fabrication process changes, such as the introduction of high-k/metal gate stack and the thin film deposition techniques, also pose difficulties for contact engineering in nano CMOS technology.

4. Contacts for 2D Materials

Two-dimensional materials, especially transition metal dichalcogenides (TMDs), such as molybdenum disulfide (MoS₂) [21–23], ditelluride (MoTe₂) [24,25], tungsten diselenide (WSe₂) [26,27], and phosphorene, are considered to be possible replacements for silicon when silicon-based CMOS devices reach their physical limits [15–20]. These materials, which have high electron mobility, low optical absorption coefficient, and high electrical and thermal conductivities, could enhance the device and circuit performance in many aspects. However, there are still unresolved challenges related to mass production, complex device design, and circuit integration [15,16].

Most of the reported 2D-material-based devices are currently much bigger than CMOS technology, even though 2D materials are often linked to nanodevices and nanotechnology in the literature. Achieving a good ohmic contact at the submicrometer scale is still a difficult problem. Because the films are only a few monolayers thick and the dopant concentration is far less than that of conventional semiconductors, the contact resistances of metal/2D materials are much larger than that of conventional semiconductors (see Figure 13). Shen et al. reported an ultralow resistance ohmic contact based on semimetallic bismuth on monolayer TMD materials. The Schottky barrier height was reduced to a zero voltage [59]. There is a large body of literature on 2D-material-based Schottky contact devices from over the last decade [60-65]. Many results have been obtained using the simple Schottky–Mott rule, neglecting the technological issues and other mechanisms related to the charge transport over the barrier, resulting in inaccurate and inappropriate results for the barrier height and pinning parameter. For example, the wide scatter of the pinning parameter versus metal workfunction plot, as shown in Figure 14, for the 2D material, provided by Wang et al. [62], is an indication of the improper treatment or interpretation of the experimental data from this simple junction. Although 2D materials have atomically flat surfaces and minimal dangling bonds or charge traps, which are advantageous for ohmic contact formation, 2D materials are difficult to dope heavily, and metal/2D material contacts are not as good as metal/Si contacts. The contact resistance is high even with a much larger surface area. These issues need to be addressed before 2Dmaterial-based devices can be scaled down to the nanoscale. The high contact resistance and the nonlinear current-voltage characteristics of the contact could limit low-voltage and high-performance circuit applications.



Figure 13. (**Top**) Plot of the contact resistance as a function of the film thickness or Si fin thickness in the case of FinFET; (**Bottom**) contact resistance as a function of the carrier density. Data are from various sources [59]. © 2021 Springer Nature. Reproduced with permission.



Figure 14. Plot of the pinning parameter S versus the metal workfunction range (ΔW_M) of various 2D materials from various sources. The labels "2L" and "3L" in brackets denote a bilayer and trilayer, respectively [62]. © 2021 Springer Nature.

Figure 15 shows the main features and challenges of metal/2D material contacts, which are summarized as follows:



Figure 15. Illustration of the issues and features associated with metal–2D material (MoS₂, as an example) contacts.

- 7. van der Waals gap: This is a tunneling barrier between the metal and the 2D material that allows for the tunneling of electrons. Some 2D materials may also form strong bonds with the metal or by overlapping their orbitals.
- 8. 2D contact: This is a common method of connecting 2D material from the top, but it has a high resistance per area because the current flows parallel to the 2D plane, not perpendicular to it. A top contact is easier to achieve because it involves depositing metal on the surface of the 2D materials and patterning it with standard photolithography.
- 9. Hybridization and edge contact: This is a better alternative to the 2D contact, as it creates a physical bond and a direct current path along the surface of the 2D material. However, it is challenging to achieve, because it requires the precise alignment and deposition of metal on very thin edges. Some 2D materials may also form strong bonds with the metal or by overlapping their orbitals.
- 10. Doping: 2D materials cannot be doped in the conventional ways, and it is hard to dope heavily.
- 11. Metal intercalation: This is a process of incorporating metal atoms into the gaps of multilayer 2D materials. The dopants contribute to the current's conduction and can improve the contact's conductivity.
- 12. Surface defects: These are imperfections, such as sulfur (S) vacancy in MoS₂, on the 2D material surface that can trap charges and pin the Fermi level, affecting the contact potential and resistance.
- 13. Layer-dependent bandgap: The bandgap and the contact potential of the 2D material vary with the number of layers. This can be exploited to tune the contact properties by changing the layer thickness in the contact region.
- 14. Phase modification: Some 2D materials can switch among different phases that have distinct electrical properties. For instance, the 2H phase of TMDs is semiconducting, while the 1T and 1T0 phases are metallic. By changing the contact region to a metallic phase, the contact conductivity can be enhanced significantly.

- 15. Buffer layer insertion: inserting a buffer layer between the metal and the 2D material can help reduce the effects of the van der Waals gap and the metal-induced gap states.
- 16. Metal workfunction selection: To achieve good ohmic contact with both n-type and p-type 2D materials, various unconventional metals, such as In, Mg, Ag, Pd, Sc, and Ti, have been explored. However, not much work has addressed the issues of stability, reliability, and potential process contamination. For digital circuit applications, one must consider whether these metals can produce the desired threshold voltages for n-type and p-type transistors.

In the remaining part of this section, we further elaborate on these issues. The recent progress on techniques to improve these 2D materials/metal contacts is highlighted. The discussion is categorized into five subsections: (1) metal van der Waals contacts and hybridization contacts; (2) junction doping; (3) bandgap modification; (4) phase modification; (5) Fermi-level pinning and buffer layers.

4.1. Metal van der Waals Contacts and Hybridized Contacts

One of the challenges in fabricating devices based on 2D materials is to ensure a good contact between the 2D lattice and the 3D metal electrodes, without damaging the 2D structure or creating interface defects. To address this issue, some researchers have developed techniques to integrate 3D metal layers with 2D materials using van der Waals (vdWs) forces. These techniques preserve the integrity of the 2D lattice and avoid chemical bonding. For example, Liu et al. [63] devised a novel method to transfer atomically smooth metal layers onto 2D semiconductors. The procedure involved patterning and depositing the metal layer on an atomically flat Si substrate, then covering it with a PMMA film to enable its detachment. Afterward, a PDMS stamp was employed to lift the PMMA-coated metal layer and align it precisely on the target surface of the 2D material. On the basis of this method, the Schottky barrier height obtained was close to the Schottky-Mott model and had a large pinning factor of 0.96 (see Figure 16). In contrast, when the metal contact was made using direct evaporation, the barrier height was much smaller, and the pinning factor was reduced to 0.09. However, the metal transfer method is not a mass production technique. For an actual integrated circuit, the surface profile will be highly uneven, and the coverage and air gaps on the step edges will be an unresolvable issue. Wang and coworkers also presented a simple technique to produce van der Waals contacts on singlelayer MoS₂ using Au-capped In as the contact material, and the contact resistance is around $3.3 \pm 0.3 \text{ k}\Omega \cdot \text{mm}$ [64].



Figure 16. Comparison of metal/MoS₂ Schottky barrier contacts produced by thermal evaporation and transferred from predeposition on atomically flat silicon surface [63]. © 2018 Springer Nature. Reproduced with permission.

From a fabrication process and device operation point of view, there should be a better approach to achieving covalent bonding between 2D materials and metals. An intimate contact would help to eliminate the interfacial tunneling barrier and enhance the carrier injection efficiency. Side or edge contacts on 2D materials are one way to obtain these kinds of metal/2D material contacts. The edge contacts can be fabricated by using either top-down or bottom-up approaches. In a top-down scheme, an insulating layer, such as Al₂O₃ or h-BN, is first deposited on the 2D channel. It is then selectively etched to expose the 2D edges. It was reported that edge-contacted graphene FETs can have a contact resistance as low as 100 Ω ·mm while maintaining a high charge carrier mobility [66]. Yang et al. also demonstrated 1D edge contacts on few-layer MoS₂ and found a high pinning factor of ~0.975 for MoS₂-based FETs with different edge contact metals [67]. The bottom-up approach involves growing 2D materials from 3D metal seeds or another 2D metal material. This method can produce high-quality and uniform edge contacts, which improve the conductivity and mobilities of TMD-based devices [68,69].

Some metal/2D material contacts are not thermally stable. The contact characteristics are affected by the interface interaction. McDonnell et al. observed a thin layer of TiO₂ when Ti was deposited on MoS₂ with a vacuum pressure of 10⁻⁶ mbar. This layer was found to have a negligible conduction band offset with MoS₂ and resulted in a low contact resistance [70]. However, when Ti was deposited in an ultrahigh vacuum of 10^{-9} mbar, the Ti atoms reacted with MoS₂, and a Ti_xS_y phase formed, which resulted in a much larger contact resistance. The formation of TiO₂ should be due to the residual oxygen or water molecules in the vacuum system or metal. The thin TiO₂ layer inhibits the direct chemical reaction of Ti and S. At an ultrahigh vacuum, the oxygen and water residuals are significantly reduced, and the direct reaction of Ti and S is possible. TixSy has poor electrical conductivity. However, Au/MoS2 contacts were also found to be dependent on the deposition pressure. English et al. found that Au contacts deposited at 10⁻⁹ torr had the lowest contact resistance of 740 Ω ·mm, which is three times smaller than those deposited at 10⁻⁶ torr [71]. Yet this difference can also be attributed to a reduced impurity absorption at the interface under high vacuum. The different surface roughnesses of the as-deposited Au films prepared by other deposition pressures should be the major cause of the different contact resistance values.

To realize CMOS devices with 2D-material-based transistors, achieving both n-type and p-type devices on a single-type 2D material and with the same metal contacts is essential. It was discovered that the polarity of multilayer MoTe₂ encapsulated in h-BN can be varied by applying thermal annealing at different temperatures [72]. Liu and coworkers found that the MoTe₂ device changed from a p-type to an n-type conduction after annealing in N₂ at 150 °C. In particularly, the electron Schottky barrier decreased to 340 meV, and the hole barrier increased to 560 meV [72]. They attributed these changes to the annealing-induced interfacial bond hybridization of the Au/MoTe₂ contacts. Hence, an interface reaction or interface hybridization could be a way to reduce the contact resistance of 2D/metal contacts. It is worth a comprehensive study.

4.2. Junction Doping

High substrate doping has been the key strategy in conventional semiconductor technology to achieve ohmic contacts in CMOS technology. Conventional semiconductors can use different doping techniques, such as ion implantation and plasma immersion ion implantation [1], to increase the doping concentration to a degeneracy level. However, these methods are not compatible with 2D materials that have only a few atomic layers. A more effective doping method for 2D materials is spontaneous charge transfer doping (SCTD) [73,74]. For p-type doping, the energy difference between the top of the valance band of the 2D material and the lowest unoccupied molecular orbital (LUMO) of the surface doping will cause electron transfer, which leads to hole accumulation and the lifting upward of the valence band in the semiconductor surface region. Similarly, suppose the bottom of the conduction band of the 2D material is below the highest occupied molecular orbital (HOMO) of the surface dopant; electrons will transfer from the HOMO of the surface doppant to the conduction band of the semiconductor and result in the electron doping. A wide range of dopants for various 2D materials are available [73]. For instance, by immersing WS₂ or MoS₂ in pure 1,2-dichloroethane (DCE) at room temperature, n-type doping with chlorine atoms was obtained [74]. The doping was due to filling sulfur vacancies with Cl in these 2D materials. With Ni metal contacts, the contact resistance values of the doped devices decrease significantly (see Figure 17) to 0.7 kΩ·mm and 0.5 kΩ·mm, respectively, for WS₂ and MoS₂. This improvement is attributed to the thinning of the Schottky barrier because of the high carrier concentration from the semiconductor's side.



Figure 17. Comparison of the Cl doping's effect on the contact resistance of WS₂ and MoS₂ contacts: (**a**) plot of the contact resistances as a function of the gap spacing between two electrodes; (**b**) schematic band diagram of the doping effects of the studied contacts. The Fermi level was pinned close to the charge neutrality level (CNL) before doping, and the Fermi level moved upward in the heavily doped 2D materials [74]. © 2014 American Chemical Society. Reproduced with permission.

Plasma doping is another method to achieve contact doping for 2D devices [75,76]. Under hydrogen plasma irradiation, Tosun et al. demonstrated that Se vacancies are produced in WSe₂ crystals [75]. As shown in Figure 18, after 12 s hydrogen plasma treatment, the E_F – E_V increased from 0.73 eV to approximately 1.19 eV, which represents a doping level of over 4×10^{17} cm⁻³ at room temperature. The near-degeneracy doping level leads to a better contact performance. For a thick device with ten monolayers of WSe₂, the contact resistance is as low as 4.4 k Ω ·µm [75]. Similarly, Kang et al. obtained ohmic contacts for p-type WSe₂ FETs with oxygen plasma treatment [76]. The O₂ plasma irradiation formed a WO₃ intermediate layer with a larger workfunction, which provided a good contact for p-type WSe₂ FETs. It was found that a focused light beam can also cause local oxidation of a MoTe₂ film [77]. The oxidized film had a higher hole concentration, improving the p-type semiconductor/metal contact region. The optical technique has the advantage of precise control, but the increased hole level is still large enough for good ohmic contact. Other nonconventional doping techniques, such as ionic liquids and polymer electrolytes [78–80], do not seem suitable for mass production.



Figure 18. Various characteristics produced by hydrogen plasma treatment of WSe₂ 2D materials: (**a**,**b**) evidence of Se reduction with a reduced Se 3d XPS peak intensity and energy shift of W 4f spectra; (**c**) Se/W ratio as a function of the plasma treatment time; (**d**) normalized valence band spectra; (**e**,**f**) change in the Fermi level and the electron doping concentration as a function of the plasma treatment's duration [75]. © 2016 American Chemical Society. Reproduced with permission.

4.3. Bandgap Modification and Band Alignment

Band alignment can be achieved by selecting a metal with a suitable workfunction. Previous studies have shown that the contact potential depends on the type of surface termination [81]. For top contacts with a monolayer of MoS₂, the Fermi-level pinning is close to the conduction band edge, and n-type Schottky barriers are formed (See Figure 19a). In this case, Al, In, and Mg may be good candidates for ohmic contacts. The edge contacts, either armchair or zigzag termination of Mo and S atoms, result in a Fermi-level pinning near the valence band, and p-type Schottky barriers are formed (see Figure 19b–d).



Figure 19. Theoretical Schottky barrier height for various metals in different contacts with monolayer MoS2: (**a**) top contact; (**b**) edge contact through connecting Mo and S atoms at an armchair termination; (**c**,**d**) edge contact through connecting Mo and S atoms at a zigzag termination, respectively [81]. © 2019 Royal Society of Chemistry. Reproduced with permission.

Two-dimensional material has a distinct nature in that the bandgap can be readily tuned by stacking a different number of 2D monolayers. The bandgaps of 2D materials vary with the number of layers and their stacking configurations, which influences the quantum confinement effects. On the basis of the first-principles calculations, Wickramaratne et al. found that the band gap of a single monolayer of hexagonal boron nitride (h-BN) is a direct bandgap semiconductor; the gap becomes indirect for multiple layers [82]. The positions of the band edges, with respect to the vacuum level, shift by 0.5 eV for the direct-to-indirect transition (see Figure 20). Thickness-dependent bandgap characteristics were also found in other 2D materials. A comprehensive study was conducted by Li and coworkers in an attempt to optimize the MoS₂ field-effect transistor's fabrication [83]. Figure 21a shows the bias dependency of area contact resistivity (Qc) extracted from the MoS₂ field-effect transistor with a different numbers of MoS2 monolayers [83]. A turn-around behavior on the number of MoS₂ monolayers was observed (see Figure 21b). This observation was explained in terms of the different junction depths for a thicker MoS₂, which affects the carrier distribution and injection path (see Figure 21c). However, it is not sufficient to explain the contact resistivity change unless the barrier height increase is considered. As shown in Figure 21e, the slope of the resistivity changes with the film thickness, and modifying the barrier height will result in a better fitting. The change in the barrier height results in different carrier injection mechanisms: thermal emission and thermal field emission resulting from the different values of the band offsets. The thickness-dependent bandgap (E_s) and the barrier height (ϕ_B) of Au/MoS₂ contacts are shown in Figure 21f, and the energy-level alignment at Au/MoS₂ interfaces for different MoS₂ thicknesses is shown in Figure 21g. The barrier height dropped from 0.65 eV to 0.33 eV, leading to a higher contact resistance, as the MoS₂ thickness grew from one monolayer to five monolayers. However, for MoS₂ thicker than five monolayers, the trend changed. The material began to act more like a bulk material than a 2D material, and the inactive layers increased the contact resistance. These results suggest a simple way to reduce the Schottky barrier by choosing 2D materials with a suitable thickness. This unique feature leads to an important processing option, which should have several potential applications for future device technology. For example, it enables bandgap and barrier engineering for devices such as Schottky diodes, photodetectors, or solar cells. It can also allow for the customization of the channel materials for field-effect transistors. Moreover, as this review discusses, it can provide a technological solution for contact engineering. This unique feature is not possible in traditional CMOS technology.



Figure 20. Variation in the bandgap and changes in the gap types of hexagonal boron nitride with different monolayers calculated from the first principle [82]. © 2018 American Chemical Society. Reproduced with permission.



Figure 21. Demonstration of the dependence of the specific contact resistivity on the MoS₂ thickness: (a) gate bias dependence; (b) thickness dependence; (c) carrier distribution and injection path; (d) carrier injection mechanisms; (e) experimental data indicating the change in the barrier height; (f) plot of the barrier height as a function of the bandgap; (g) illustration of the energy-level alignment at Au/MoS₂ interfaces for different MoS₂ thicknesses [83]. © 2014 American Chemical Society. Reproduced with permission.

4.4. Phase Modification

Two-dimensional materials can exist in different crystal structures or phases. Different phases have different electronic and optical properties and, thus, different metal contact behaviors as well [84,85]. For example, 2H-TMDs are semiconducting materials with a trigonal prismatic structure, while 1T and 1T0 phases are metallic materials with an octahedral structure. Using n-butyllithium to induce a phase transition from 2H to 1T' or 1T0 in a MoS₂ nanosheet, Kappera et al. achieved a significant reduction in contact resistance to 0.24 k Ω ·µm [85]. Sun et al. also reported similar results with a 1T' phase contact in MoS₂ [86]. In contrast, molybdenum ditelluride (MoTe₂) has a very small energy difference (~35 meV) between the 2H and 1T0 phases in a single layer, which enables a low barrier metal/MoTe₂ contact with the 1T0 phase [87]. Qi et al. demonstrated an ohmic contact with a very low Au/MoTe2 barrier height of 10 meV with a thermal anneal-induced phase transition [88]. Using a laser-induced phase transition, the metallic monoclinic 1T' phase of MoTe₂ was obtained from the 2H phase MoTe₂ by Cho et al. [89]. It was reported that the barrier can be reduced from 0.2 eV to 0.01 eV by changing the 2H contact to a 1T' contact [89]. The significantly reduced Schottky barrier leads to high conductivity and an increase in the carrier mobility of the MoTe₂ transistor by 50 times [89].

4.5. Fermi-Level Pinning and Buffer Layer

The interface states at two-dimensional material/metal interfaces can affect the performance of two-dimensional devices. These states include metal-induced gap states (MIGS) and defect states, which can induce Fermi-level pinning (FLP) at the interface and hinder the carrier injection efficiency. Fermi-level pinning causes adverse effects in 2D material/metal contacts. It was proposed to be one of the main reasons for the experimental observations of large contact resistance. The origins of the FLP effect at the 2D material/metal interface should be due to several causes. As mentioned, the barrier height of a metal/semiconductor contact can be tailored by the proper choice of metal with the desired workfunctions according to the Schottky–Mott rule. However, the principle is not always valid because of the Fermi-level pinning effect. Similar to Fermi-level pinning in a high-defect Si/metal oxide interface, intrinsic defects or impurities, such as sulfur vacancies or surface metal-like impurities on TMD materials, can act as electron donors [90]. In addition, various physical or chemical processing steps during a device's fabrication can also produce various types of damage to the crystalline structure of 2D materials or trigger interface chemical reactions, which are additional regimes for the Fermi-level pinning of a contact.

Fermi-level pinning is often considered to arise from metal-induced gap states (MIGS) as a result of the metal electronic wave function overlapping with the 2D material at the charge neutrality level of the semiconductor. Guo et al. [91] computed the Schottky barrier of various metals on transition metal dichalcogenides (TMDs) without intrinsic defects and found a strong pinning effect with a pinning factor of S = 0.3. They attributed this result to the strong chemical bond between metals and chalcogens [91]. This type of FLP effect was verified by scanning tunneling microscopy (STM) experiments and theoretical models [92]. Kerelsky et al. showed the existence of MIGS in MoS₂ using scanning tunneling spectroscopy (STS) to measure the local density of states (LDOS) along the 2D/metal interface. The LDOS gap decreases as the contact edge becomes closer [92].

A buffer layer can reduce the effects of metal-induced gap states (MIGS) by creating a matching layer between the 2D material and the metal. This idea was inspired by previous work on high-performance metal-insulator-semiconductor (MIS) diodes [33,42]. The buffer layer prevents the metal wave function from penetrating the semiconductor material, which lowers the MIGS density. Moreover, if the buffer layer is an insulator, it can balance the charge at the interface and shift the Fermi-level closer to the charge neutrality level, which further decreases the effective Schottky barrier height. However, as given in Figure 7, the buffer layer should be thin enough to allow for efficient tunneling and have good interface quality with both the 2D material and the metal. The buffer layer can be metallic, semiconducting, or insulating.

One of the most often used buffer materials for 2D material/metal contacts is graphene. Graphene has a semi-metallic nature and a tunable workfunction. By inserting prepatterned single-layer graphene between MoS2 and metal, Chee and co-workers created a sandwich contact structure for MoS₂ FETs [93]. The graphene layer enhanced the contact between MoS₂ and Ag significantly. The Schottky barrier height was reduced to 190 meV due to the charge transfer from Ag to graphene that matched the Fermi level of graphene with a conduction band edge of MoS2. This increased the electron mobility by almost three times. Graphene is not the only metallic 2D material that can act as an interfacial buffer layer to achieve ohmic contacts for 2D-material-based devices. Other examples include NbS₂, WTe₂, and heavily p-doped Wse₂ or MoS₂ [94–96]. Metal oxides, such as TiO₂, Ta₂O₅, MgO, and ZnO, are also suitable for metal/insulator/2D semiconductor structures [97–99]. A buffer layer thickness dependence study was conducted for the MoS₂/Ta₂O₅/Ti structure. By inserting a Ta₂O₅ layer with different thicknesses up to 5 nm for the MoS₂/metal contact, Lee et al. [99] found that the contact resistance first decreases as the Ta₂O₅ buffer layer increases (see Figure 22). For increases in the buffer layer thickness to 2 nm, the contact resistance increased (see Figure 22g). The conductivity improvement is explained by the de-pinning effect, as shown in Figure 22e, and the barrier-lowering effect due to the interface dipole (see Figure 22f). When the Ta_2O_5 thickness was greater than 2 nm, the contact resistance also increased as the tunneling probability decreased with the thicker buffer layer. Therefore, there is an optimal range of the buffer layer thickness for minimizing the contact resistance of the MoS₂/Ta₂O₅/Ti structure. Many previous studies have neglected the bandgap or band offset between the insulator and semiconductor in their analysis. However, as discussed in Section 2, in relation to (8), these factors also affect the tunneling probability at the semiconductor/insulator interface. Therefore, they should be considered important parameters for optimizing the process.



Figure 22. Effects of the Ta₂O₅ buffer layer on the contact characteristics of Au-Ti/MoS₂ contacts: (**a**,**b**) device structure, layout, and scales; (**c**) resistance value as a function of the contact separation; (**d**–**f**) explanation of the Fermi-level pinning and the impact of the Ta₂O₅ buffer layer; (**g**) plot of the measured specific contact resistivity, ρ_c , as a function of the Ta₂O₅ dielectric thickness [99]. © 2016 American Chemical Society. Reproduced with permission.

Shen et al. recently discovered that a Bi semimetal layer sandwiched between the MoS₂ and Au electrode can effectively lower the MIGS density and the contact resistance [59]. This is because the Bi Fermi level is slightly above the conduction band edge of MoS₂, resulting in a negligible barrier height and a high potential for achieving good ohmic contacts. Moreover, the density functional theory (DFT) calculation shows that the p_z orbital of Bi resonates with the p_z and d_z^2 orbitals of MoS₂, shifting the Fermi level of MoS₂ into the conduction band and creating a degenerate state of MoS₂ (see Figure 23a). In this situation, the MIGS are fully saturated with electrons, and as the valence band states decrease faster than the MIGS increase (see Figure 23b), this, in effect, reduces the MIGS level [59].



Figure 23. Density functional theory calculation results of the projected local density of states (PLDOS) of the MoS₂ (top), MoS₂ in contact with Bi (middle), Bi (bottom); (**b**) change in PLDOS of different orbitals in the valence band and MIGS region [59]. © 2021 Springer Nature. Reproduced with permission.

One of the challenges in fabricating MIS devices based on 2D materials is the difficulty of growing a uniform oxide layer on 2D materials due to the lack of evenly distributed surface dangling bonds [100,101]. A possible solution to this challenge is to use hexagonal boron nitride (h-BN) as an interlayer. Wang et al. [91] demonstrated that inserting one or two layers of h-BN between Ni and MoS₂ can significantly reduce the Schottky barrier from 158 meV to 31 meV and, consequently, lower the contact resistance from 5.1 k Ω ·mm to 1.8 k Ω ·mm. Similarly, by inserting a h-BN layer between MoTe₂ and Sc, a highperformance n-type MoTe₂ FET was achieved [101]. One way to gain better control of the threshold voltage and reliability in CMOS technology is to use surface or dangling bond passivation. This technique can also be applied to 2D materials. Cho et al. demonstrated chemisorbed thiol molecules as a tunneling layer at the interface between MoS₂ and metal [102]. Thiol molecules can eliminate interface states that result from sulfur vacancies and create additional tunneling channels for charge injection through field emission. This lowers the interface barrier and enables ohmic contacts.

5. 2D Contact with Silicon

The current 2D-material-based electron devices are still much larger than state-ofthe-art CMOS technology. There is a lack of mass production and large-scale integration technology for these devices. Therefore, it is unlikely that 2D-material-based ICs will replace the mainstream silicon technology in the next ten years. However, it is highly possible that some 2D materials will be integrated with Si technology to overcome some of the limitations of CMOS devices and fabrication technology and to enhance the performance of silicon devices. For instance, using 2D materials instead of the expensive indium tin oxide (ITO) films in photonic devices could be a promising option for 2D materials/Si technology integration. Graphene–silicon solar cells have been extensively studied since the discovery of graphene [103–105]. However, some fundamental issues of the graphene/silicon interface are still not well understood. Some puzzling issues, such as the wide variations in Schottky barrier height and large fluctuations in the ideality factor (from ~1 to 30), were observed in this simple structure [41,96]. These wide ranges in parameter values suggest that some additional physical mechanisms in addition to the Schottky emission should be considered.

The interface between 2D material and silicon has a unique nature that is not common in the conventional silicon process and is also unknown to the 2D material community [41]. Figure 23 shows the different types of material interfaces [41]. Two-dimensional materials, such as graphene, have a much lower surface defect density than most conventional semiconductor materials. However, silicon has many dangling bonds on its surface. The van der Waals contact of the graphene/Si interface would leave a large number of unpassivated silicon dangling bonds. Figure 24a depicts a freshly etched <100> silicon surface full of silicon dangling bonds, also called P_{b0} centers or denoted as ≡Si. [106–108]. A Pb0 center can capture both electrons and holes. It is an amphoteric defect center. Thus, it would contribute to the current conduction in both forward and reverse biases. In MOS transistors or Si technology, defects can be readily passivated with well-developed forming gas annealing. In the forming gas annealed Si surface or Si/SiO2 interface, the defects are passivated with a hydrogen- or water-related hydroxyl group (see Figure 24b). The defect density can be greatly reduced If the surface is oxidized (see Figure 24c) or bonded to other materials, such as metal (see Figure 24d). Figure 24e,f illustrate the graphene/Si contact. Graphene may be able to form covalent bonds, such as in Gr/SiC structures and edge contact, as mentioned in Section 4.1. For 2D van der Waals contacts, there is no strong physical bonding, either covalent or ionic, between the silicon and the 2D graphene layer. This is a distinct contact that cannot be found in conventional semiconductor contacts. Because the silicon dangling bonds are unpassivated, the number of defects is significantly higher than that of other interfaces. This results in significant Fermi-level pinning. However, as the hexagonal openings in the graphene lattice are so small that most of the atoms or molecules are too large to pass through them, the defects are physically isolated from other materials. This model was used to explain the unusual current-voltage characteristics of graphene/Si contacts [41].



Figure 24. Six different surfaces/contacts with Si: (**a**) bare silicon surface and surface dangling bonds or Pb0 centers; (**b**) Pb0 centers passivated with hydrogen and hydroxyl; (**c**) silicon covered by silicon oxide; (**d**) metal/Si contact; (**e**) graphene/Si contact; (**f**) Si/carbon covalent contact [41]. © 2021 Elsevier. Reproduced with permission.

Figure 25 shows the forward current-voltage characteristics of 2D material/Si Schottky junctions from different sources [41,100]. We can see that these I-V curves do not follow the expected Schottky behavior. The Schottky equations do not fit the current well, and the ideality factors from the fitting are very large and inconsistent. This may be due to the oxide layers or interface defects in the junctions [41]. These factors can affect the charge transport mechanisms, such as thermionic emission, thermionic field emission, or tunneling, and increase the junction current, especially at high bias voltages. Figure 26 illustrates how Pb0 centers can contribute to the current's conduction under both forward and reverse biases [41]. The main current under forward bias is caused by the thermionic emission of electrons from the silicon conduction band to the graphene conduction band via mechanism (1). Some of these electrons may be trapped by the P_{b0} centers, as shown in mechanism (2). The trapped electrons may be emitted into the conduction band of graphene via mechanism (3). These charge transport processes can result in abnormal values of the effective barrier height and ideality factor if one fits the I-V characteristics with the Schottky equation as provided in (7) or (8). Additional physical mechanisms in addition to the Schottky emission should be considered, which also implies possible techniques for improvements in the contact characteristics.



Figure 25. Abnormal current–voltage characteristics of a graphene/n-type Si Schottky diode taken from Ref. [41] (a) and Ref. [109] (b). © 2021, 2022, Elsevier. Reproduced with permission.



Figure 26. The involvement of silicon Pb0 centers in the current conduction of graphene/Si junctions: (a) equilibrium; (b) forward bias; (c) reverse bias [41]. © 2021 Elsevier. Reproduced with permission.

In summary, the interface between 2D materials and silicon is different than other interfaces. The contact behavior and the contact stability depend mainly on the surface properties of the silicon, especially the surface defects and how they are treated. The Schottky equation is not very accurate for describing the current flow in this case.

6. Concluding Remarks

As CMOS technology further shrinks to the nanoscale range in the coming technological nodes, aggressive contact scaling is indispensable. This scaling will lead to some poor contacts because of the more significant impact of the surface roughness, contact size reduction, and the nonscalability of film interfaces, as well as the use of undoped substrates. These factors lead to higher contact resistance and larger characteristic nonlinearity, which could hinder performance improvement in further-scaled CMOS devices and integrated circuits. On the other hand, although the emerging two-dimensional materials with smooth surfaces and dangling-bond-free nature have the potential to offer better material interfaces and form good electrical contacts, achieving signal transparent ohmic contacts is still a big challenge. The contacts between two-dimensional materials and semiconductors, as well as between two-dimensional materials and semischottky barriers, and they do not follow the Schottky–Mott rule due to surface defects pinning the Fermi level and the van der Waals nature of the material interfaces.

In this review, we examined some of the theoretical background of metal/semiconductor contacts and highlight the validity of the Schottky equation. Then, we discussed the issues associated with reducing the contact size and the effects of a nanometer CMOS device's structure and fabrication technique on ohmic contacts. We provided a systematic survey of the recent advancements and technological trends in contact engineering, such as phase and bandgap engineering, 2D/metal van der Waals contacts and hybrid contacts, junction doping technology, contacts with a buffer layer, for emerging 2D materials. This technological overview should offer new insight and solutions for overcoming contact scaling issues and facilitates the technological development for abridging the 2D materials with future CMOS technology.

Author Contributions: H.W. developed the concepts, performed the literature search, analyzed the data, developed the theories, and wrote the article. J.Z. and J.L. provided funding support, developed the concepts, and proofread the article. All authors have read and agreed to the published version of the manuscript.

Funding: This work was supported by project #9239120 of City University of Hong Kong, Hong Kong SAR, China, which was funded by Hubei JFS Lab, Wuhan, China.

Conflicts of Interest: The authors declare no conflicts of interest.

Nomenclature

2D	Two Dimensional
3D	Three Dimensional
ALD	Atomic Layer Deposition
AM	Amplitude Modulation
BN	Boron Nitride
CFET	Complementary Field-Effect Transistor
CGP	Contacted Gate Pitch
CMOS	Complementary Metal-Metal-Semiconductor
CPP	Contact-Poly Pitch
DCE	Dichloroethane
DFT	Density Functional Theory
FET	Field-Effect Transistor
FinFET	Fin Field-Effect Transistor
FLP	Fermi-Level Pinning
FN	Fowler–Nordheim
GAA	Gate-All-Around
GR	Generation and Recombination
HOMO	Highest-Occupied Molecular Orbital
IC	Integrated Circuit
ITRS	International Technology Roadmap for Semiconductors
I-V	Current-Voltage (Characteristics)
LDOS	Local Density of States
LED	Light-Emitting Diode
LUMO	Lowest Unoccupied Molecular Orbital
MIGS	Metal-Induced Gap States
MIM	Metal–Insulator–Metal
MIS	Metal-Insulator-Semiconductor
MOS	Metal-Oxide-Metal-Semiconductor
PDMS	Polydimethylsiloxane
PF	Poole-Frenkel (emission)
PLDOS	Projected Local Density of States
PMMA	Polymethylmethacrylate
SCTD	Spontaneous Charge Transfer Doping
STM	Scanning Tunneling Microscopy
STS	Scanning Tunneling Spectroscopy
TMD	Transition Metal Dichalcogenide
TSV	Through Silicon Via

References

1. Iwai, H.; Sze, S.M.; Taur, Y.; Wong, H. MOSFETs. In *Guide to State-of-the-Art Electron Devices*; Burghartz, J.N. Ed.; Wiley and IEEE: Hoboken, NJ, USA, 2013; pp. 21–36.

2. Moore, G.E. Cramming More Components onto Integrated Circuits. *Electronics* 1965, 38, 114–117.

3. Wong, H.; Iwai, H. The road to miniaturization. Phys. World 2005, 18, 40.

4. Dennard, R.H.; Gaensslen, F.; Yu, H.-N.; Rideout, L.; Bassous, E.; LeBlanc, A. Design of ion-implanted MOSFET's with very small physical dimensions. *IEEE J. Solid-State Circuits* **1974**, *9*, 256–268.

5. Semiconductor Industry Association, 2013 International Technology Roadmap for Semiconductors. Available online: https://www.semiconductors.org/resources/2013-international-technology-roadmap-for-semiconductors-itrs/ (accessed on December 30, 2019).

6. Wong, H.; Iwai, H. On the Scaling of Subnanometer EOT Gate Dielectrics for Ultimate nano CMOS Technology. *Microelectron. Eng.* **2015**, *138*, 57–76.

7. Wong, H. On the CMOS Device Downsizing, More Moore, More than Moore, and More-than-Moore for More Moore. In Proceedings of the 2021 IEEE 32nd International Conference on Microelectronics (MIEL), Nis, Serbia, 12–14 September 2021; pp. 9–15.

8. Wong, H.; Kakushima, K. Poole–Frenkel (PF)-MOS: A Proposal for the Ultimate Scale of an MOS Transistor. *Nanomaterials* **2023**, *13*, 411.

9. Ye, P.D.; Ernst, T.; Khare, M.V. the Nanosheet Transistor Is the Next and Maybe Last Step in Moores-Law. *IEEE Spectr.* **2019**, *30*. https://spectrum.ieee.org/the-nanosheet-transistor-is-the-next-and-maybe-last-step-in-moores-law

10. Wong, H.; Kakushima, K. On the Vertically Stacked Gate-All-Around Nanosheet and Nanowire Transistor Scaling beyond the 5 nm Technology Node. *Nanomaterials* **2022**, *12*, 1739.

Li, C.; Pan, Z.; Li, X.; Hao, W.; Miao, R.; Wang, A. Selective Overview of 3D Heterogeneity in CMOS. *Nanomaterials* 2022, *12*, 2340.
 Razavieh, A.; Zeitzoff, P.; Brown, D.E.; Karve, G.; Nowak, E.J. Scaling Challenges of FinFET Architecture below 40nm Contacted Gate Pitch. In Proceedings of the 2017 75th Annual Device Research Conference (DRC), South Bend, IN, USA, 25–28 June 2017; pp. 1–2.

13. Afzalian, A.; Ahmed, Z.; Ryckaert, J. Towards Sub-30nm Contacted Gate Pitch, Forked Contact and Dynamically-Doped Nanosheets to Enhance Si and 2D Materials Device Scaling. *arXiv* **2022**, arXiv:2203.06364. https://doi.org/10.48550/arXiv.2203.06364.

14. Liu, Y.; Duan, X.; Shin, H.J.; Park, S.; Huang, Y.; Duan, X. Promises and Prospects of Two-Dimensional Transistors. *Nature* **2021**, *591*, 43–53.

15. Wong, H. Abridging CMOS Technology. Nanomaterials 2022, 12, 4245.

16. Elmourabit, F.; Dlimi, S.; El Moutaouakil, A.; Id Ouissaaden, F.; Khoukh, A.; Limouny, L.; Elkhatat, H.; El Kaaouachi, A. Nature of the Metal Insulator Transition in High-Mobility 2D Si-MOSFETs. *Nanomaterials* **2023**, *13*, 2047.

17. Knobloch, T.; Selberherr, S.; Grasser, T. Challenges for Nanoscale CMOS Logic based on Two-Dimensional Materials. *Nanomaterials* **2022**, *12*, 3548.

18. Yu, J.; Wu, S.; Zhao, X.; Li, Z.; Yang, X.; Shen, Q.; Lu, M.; Xie, X.; Zhan, D.; Yan, J. Progress on Two-Dimensional Transitional Metal Dichalcogenides Alloy Materials: Growth, Characterisation, and Optoelectronic Applications. *Nanomaterials* **2023**, *13*, 2843.

19. Filipovic, L.; Selberherr, S. Application of Two-Dimensional Materials towards CMOS-Integrated Gas Sensors. *Nanomaterials* **2022**, *12*, 3651.

20. Kizhepat, S.; Rasal, A.S.; Chang, J.-Y.; Wu, H.-F. Development of Two-Dimensional Functional Nanomaterials for Biosensor Applications: Opportunities, Challenges, and Future Prospects. *Nanomaterials* **2023**, *13*, 1520.

21. Lemme, M.C.; Daus, A. Low-temperature MoS₂ Growth on CMOS Wafers. Nat. Nanotechnol. 2023, 18, 446–447.

22. Pelella, A.; Kharsah, O.; Grillo, A.; Urban, F.; Passacantando, M.; Giubileo, F.; Iemmo, L.; Sleziona, S.; Pollmann, E.; Madauß, L.; et al. Electron Irradiation of Metal Contacts in Monolayer MoS₂ Field-Effect Transistors. *ACS Appl. Mater. Interfaces* **2020**, *12*, 40532–40540.

23. Di Bartolomeo, A.; Kumar, A.; Durante, O.; Sessa, A.; Faella, E.; Viscardi, L.; Intonti, K.; Giubileo, F.; Martucciello, N.; Romano, P.; et al. Temperature-dependent Photoconductivity in Two-dimensional MoS₂ Transistors. *Mater. Today Nano* **2023**, *24*, 100382.

24. Chen, J.; Zhu, J.; Wang, Q.; Wan, J.; Liu, R. Homogeneous 2D MoTe₂ CMOS Inverters and p-n Junctions Formed by Laser-Irradiation-Induced p-Type Doping. *Small* **2020**, *16*, 2001428.

25. Lim, J.Y.; Pezeshki, A.; Oh, S.; Kim, J.S.; Lee, Y.T.; Yu, S.; Hwang, D.K.; Lee, G.H.; Choi, H.J.; Im, S. Homogeneous 2D MoTe₂ p–n Junctions and CMOS Inverters Formed by Atomic-Layer-Deposition-Induced Doping. *Adv. Mater.* **2017**, *29*, 1701798.

26. Patoary, N.H.; Xie, J.; Zhou, G.; Al Mamun, F.; Sayyad, M.; Tongay, S.; Esqueda, I.S. Improvements in 2D p-type WSe₂ Transistors Towards Ultimate CMOS Scaling. *Sci. Rep.* **2023**, *13*, 3304.

27. Yu, L.; Zubair, A.; Santos, E.J.G.; Zhang, X.; Lin, Y.; Zhang, Y.; Palacios, T. High-performance WSe₂ Complementary Metal Xxide Semiconductor Technology and Integrated Circuits. *Nano Lett.* **2015**, *15*, 4928–4934.

- 28. Grondahl, L.O. Twenty-Five Years of Copper-Copper Oxide Rectifiers. Trans. Am. Inst. Electr. Eng. 1948, 67 Pt I, 403-410.
- 29. Nordheim, L. Zur Theorie der Detektorwirkung. Z. Für Phys. 1932, 75, 434-441.

30. Frenkel, J.; Joffe, A. On the Electric and Photoelectric Properties of Contacts between a Metal and a Semiconductor. *Phys. Rev.* **1932**, 39, 530–531.

- 31. Fowler, R.H. Statistical Mechanics; Cambridge University Press: Cambridge, UK, 1936.
- 32. Schottky, W. Zur Halbleitertheorie der Sperrschicht- und Spitzengleichrichter. Z. Physik 1939, **113**, 367–414. https://doi.org/10.1007/BF01340116
- 33. Mott, N.F. Note on the Contact between a Metal and an Insulator or Semiconductor. *Math. Proc. Camb. Philos. Soc.* **1938**, 34, 568–572.

34. Richardson, O.W. On the Negative Radiation from Hot Platinum. In *Proceedings of the Philosophical of the Cambridge Philosophical Society;* Cambridge University Press: Cambridge, UK, 1901; Volume 11, pp. 286–295.

- 35. Bardeen, J. Surface States and Rectification at a Metal Semi-conductor Contact. Phys. Rev. 1947, 71, 717–727.
- 36. Tung, R.T. Schottky Barrier Height-Do We Really Understand What We Measure? J. Vac. Sci. Technol. B 1993, 11, 1546–1552.
- 37. Tung, R.T. the physics and Chemistry of the Schottky Barrier Height. Appl. Phys. Rev. 2014, 1, 011304.
- 38. Crowell, C.R. the Richardson Constant for Thermionic Emission in Schottky Barrier Diodes. Solid-State Electron. 1965, 8, 395–399.

39. Shockley, W. The Theory of *p*-*n* Junctions in Semiconductors and *p*-*n* Junction Transistors. *Bell Syst. Tech. J.* **1949**, *28*, 435–489.

40. Schenk, A.; Krumbein, U. Coupled Defect-Level Recombination: Theory and Application to Anomalous Diode Characteristics. *J. Appl. Phys.* **1995**, *78*, 3185–3192.

41. Wong, H.; Anwar, M.A.; Dong, S. Effects of Silicon Surface Defects on the Graphene/Silicon Schottky Characteristics. *Results Phys.* **2021**, 29, 104744.

42. Card, H.C.; Rhoderick, E.H. Studies of Tunnel MOS Diodes I. Interface Effects in Silicon Schottky Diodes. J. Phys. D Appl. Phys. **1971**, *4*, 1589–1601.

43. Ando, T.; Fowler, A.B.; Stern, F. Electronic Properties of Two-dimensional Systems. Rev. Mod. Phys. 1982, 54, 437.

44. Wong, H.; Iwai, H. Modeling and Characterization of Direct-Tunneling Current in Dual-Layer Ultrathin-Gate Dielectric Films. *J. Vac. Sci. Technol. B* **2006**, *24*, 1785–1793.

45. Yang, B.L.; Lai, P.T.; Wong, H. Conduction Mechanisms in MOS Gate Dielectric Films. Microelectron. Reliab. 2004, 44, 709–718.

46. Toyama, N. Variation in the Effective Richardson Constant of a Metal-Silicon Contact due to metal Film Thickness. *J. Appl. Phys.* **1988**, *63*, 2720–2724.

47. Yu, A. Electron Tunneling and Contact Resistance of Metal-Silicon Contact Barriers. Solid-State Electron. 1970, 13, 239–247.

48. Smoluchowski, R. Anisotropy of the Electronic Work Function of Metals. Phys. Rev. 1941, 60, 661–674.

49. Zhao, Y.P.; Wang, G.C.; Lu, T.M.; Palasantzas, G.; de Hosson, J.T.M. Surface-Roughness Effect on Capacitance and Leakage Current of An Insulating Film. *Phys. Rev. B* **1999**, *60*, 9157–9164.

50. Wong, H.; Zhang, J.; Iwai, H.; Kakushima, K. Characteristic Variabilities of Subnanometer EOT La₂O₃ Gate Dielectric Film of Nano CMOS Devices. *Nanomaterials* **2021**, *11*, 2118.

51. Alimardani, N.; Conwell, E.W.; Wager, J.F.; Conley, J.F.; Evans, D.R.; Chin, M.; Kilpatrick, S.J.; Dubey, M. Impact of Electrode Roughness on Metal-Insulator-Metal Tunnel Diodes with Atomic Layer Deposited Al₂O₃ Tunnel Barriers. *J. Vac. Sci. Technol. A* 2012, 30, 01A113.

52. Du, L.; Wong, H.; Dong, S.; Lau, W.S.; Filip, V. AFM Study on the Surface Morphologies of Tin Films Prepared by Magnetron Sputtering and Al2O3 Films Prepared by Atomic Layer Deposition. *Vacuum* **2018**, *153*, 139–144.

53. Yu, D.Q.; Lau, W.S.; Wong, H.; Feng, X.; Dong, S.; Pey, K.L. the Variation of the Leakage Current Characteristics of W/Ta₂O₅/W MIM Capacitors with the Thickness of the Bottom W Electrode. *Microelectron. Reliab.* **2016**, *61*, 95–98.

54. Lau, W.S.; Zhang, J.; Wan, X.; Luo, J.K.; Xu, Y.; Wong, H. Surface Smoothing Effect of An Amorphous Thin Film Deposited by Atomic Layer Deposition on a Surface with Nano-Sized Roughness. *AIP Adv.* **2014**, *4*, 027120.

55. Liu, Y.; Kijima, S.; Sugimata, E.; Masahara, M.; Endo, K.; Matsukawa, T. Investigation of the Tin Gate Electrode with Tunable Work Function and Its Application for Finfet Fabrication. *IEEE Trans. Nanotechnol.* **2006**, *5*, 723–730.

56. Zheng, P.Y.; Zhou, T.; Engler, B.J.; Chawla, J.S.; Hull, R.; Gall, D. Surface Roughness Dependence of the Electrical Resistivity of W(001) Layers. *J. Appl. Phys.* **2017**, *122*, 095304.

57. Sami, S.N.; Diaz, L.; Sanati, M.; Joshi, R.P. Simulations of Field Emission from Copper Electrodes with Inclusion of Oxygen Surface Layer and Work Function Changes based on First-Principles Calculations. *J. Appl. Phys.* **2020**, *128*, 223302.

58. Li, Z.; Rahtu, A.; and Gordon, R.G. Atomic Layer Deposition of Ultrathin Copper Metal Films from a Liquid Copper(I) Amidinate Precursor. J. Electrochem. Soc. 2006, 153, C787–C794.

59. Shen, P.C.; Su, C.; Lin, Y.; Chou, A.S.; Cheng, C.C.; Park, J.H.; Chiu, M.H.; Lu, A.Y.; Tang, H.L.; Tavakoli, M.M.; et al. Ultralow Contact Resistance between Semimetal and Monolayer Semiconductors. *Nature* **2021**, *593*, 211–217.

60. Zheng, Y.; Gao, J.; Han, C.; Chen, W. Ohmic Contact Engineering for Two-Dimensional Materials. *Cell Rep. Phys. Sci.* 2021, 2, 100298.

61. Schulman, D.S.; Arnold, A.J.; Das, S. Contact Engineering for 2D Materials and Devices. Chem. Soc. Rev. 2018, 47, 3037–3058.

62. Wang, Q.; Cao, L.; Liang, S.-J.; Wu, W.; Wang, G.; Lee, C.H.; Ong, W.L.; Yang, H.Y.; Ang, L.K.; Yang, S.A.; et al. Efficient Ohmic Contacts and Built-In Atomic Sublayer Protection in MoSi₂N₄ and WSi₂N₄ Monolayers. *npj* 2D Mater. Appl. **2021**, *5*, 71.

63. Liu, Y.; Guo, J.; Zhu, E.; Liao, L.; Lee, S.-J.; Ding, M.; Shakir, I.; Gambin, V.; Huang, Y.; Duan, X. Approaching the Schottky–Mott Limit in Van Der Waals Metal–Semiconductor Junctions. *Nature* **2018**, *557*, 696–700.

64. Wang, Y.; Kim, J.C.; Wu, R.J.; Martinez, J.; Song, X.; Yang, J.; Zhao, F.; Mkhoyan, A.; Jeong, H.Y.; Chhowalla, M. Van Der Waals Contacts between Three-Dimensional Metals and Two-Dimensional Semiconductors. *Nature* **2019**, *568*, 70–74.

65. Grillo, A.; DiBartolomeo, A. A Current–Voltage Model for Double Schottky Barrier Devices. Adv. Electron. Mater. 2021, 7, 2000979.

66. Wang, L.; Meric, I.; Huang, P.Y.; Gao, Q.; Gao, Y.; Tran, H.; Taniguchi, T.; Watanabe, K.; Campos, L.M.; Muller, D.A.; et al. One-Dimensional Electrical Contact to a Two-Dimensional Material. *Science* **2013**, *342*, 614–617.

67. Yang, Z.; Kim, C.; Lee, K.Y.; Lee, M.; Appalakondaiah, S.; Ra, C.H.; Watanabe, K.; Taniguchi, T.; Cho, K.; Hwang, E.; et al. A Fermi-Level-Pinning-Free 1D Electrical Contact at the Intrinsic 2D MoS₂-Metal Junction. *Adv. Mater.* **2019**, *31*, e1808231.

68. Chu, C.H.; Lin, H.C.; Yeh, C.H.; Liang, Z.Y.; Chou, M.Y.; Chiu, P.W. Endbonded Metal Contacts on WSe₂ Field-Effect Transistors. ACS Nano **2019**, *13*, 8146–8154.

69. Guimaraes, M.H.D.; Gao, H.; Han, Y.; Kang, K.; Xie, S.; Kim, C.J.; Muller, D.A.; Ralph, D.C.; Park, J. Atomically Thin Ohmic Edge Contacts between Two-Dimensional Materials. ACS Nano 2016, 10, 6392–6399.

70. McDonnell, S.; Smyth, C.; Hinkle, C.L.; Wallace, R.M. MoS₂–Titanium Contact Interface Reactions. *ACS Appl. Mater. Interfaces* **2016**, *8*, 8289–8294.

71. English, C.D.; Shine, G.; Dorgan, V.E.; Saraswat, K.C.; Pop, E. Improved Contacts to MoS₂ Transistors by Ultra-High Vacuum Metal Deposition. *Nano Lett.* **2016**, *16*, 3824–3830.

72. Liu, X.; Islam, A.; Guo, J.; Feng, P.X.-L. Controlling Polarity of MoTe² Transistors for Monolithic Complementary Logic Via Schottky Contact Engineering. ACS Nano 2020, 14, 1457–1467.

73. Zhang, X.; Shao, Z.; Zhang, X.; He, Y.; Jie, J. Surface Charge Transfer Doping of Low-Dimensional Nanostructures Toward High-Performance Nanodevices. *Adv. Mater.* **2016**, *28*, 10409–10442.

74. Yang, L.; Majumdar, K.; Liu, H.; Du, Y.; Wu, H.; Hatzistergos, M.; Hung, P.Y.; Tieckelmann, R.; Tsai, W.; Hobbs, C.; et al. Chloride Molecular Doping Technique on 2D Materials: WS₂ and MoS₂. *Nano Lett.* **2014**, *14*, 6275–6280.

75. Tosun, M.; Chan, L.; Amani, M.; Roy, T.; Ahn, G.H.; Taheri, P.; Carraro, C.; Ager, J.W.; Maboudian, R.; Javey, A. Airstable N-Doping of Wse2 by Anion Vacancy Formation with Mild Plasma Treatment. *ACS Nano* **2016**, *10*, 6853–6860.

76. Kang, W.M.; Lee, S.T.; Cho, I.T.; Park, T.H.; Shin, H.; Hwang, C.S.; Lee, C.; Park, B.G.; Lee, J.H. Multi-Layer WSe₂ Field Effect Transistor with Improved Carrier-Injection Contact by using Oxygen Plasma Treatment. *Solid-State Electron*. **2018**, *140*, 2–7.

77. Seo, S.Y.; Park, J.; Park, J.; Song, K.; Cha, S.; Sim, S.; Choi, S.Y.; Yeom, H.W.; Choi, H.; Jo, M.H. Writing Monolithic Integrated Circuits on a Two-Dimensional Semiconductor with a Scanning Light Probe. *Nat. Electron.* **2018**, *1*, 512–517.

78. Liu, X.; Qu, D.; Ryu, J.; Ahmed, F.; Yang, Z.; Lee, D.; Yoo, W.J. P-Type Polar Transition of Chemically Doped Multilayer MoS² Transistor. *Adv. Mater.* **2016**, *28*, 2345–2351.

79. Kiriya, D.; Tosun, M.; Zhao, P.; Kang, J.S.; Javey, A. Air-Stable Surface Charge Transfer Doping of MoS₂ by Benzyl Viologen. *J. Am. Chem. Soc.* **2014**, *136*, 7853–7856.

80. Yue, D.; Kim, C.; Lee, K.Y.; Yoo, W.J. Ohmic Contact in 2D Semiconductors via the Formation of a Benzyl Viologen Interlayer. *Adv. Funct. Mater.* **2019**, *29*, 1807338.

81. Fang, Q.; Zhao, X.; Huang, Y.; Xu, K.; Min, T.; Ma, F. Junction-Configuration-Dependent Interfacial Electronic States of a Monolayer Mos₂/Metal Contact. J. Mater. Chem. C 2019, 7, 3607–3616.

82. Wickramaratne, D.; Weston, L.; Van de Walle, C.G. Monolayer to Bulk Properties of Hexagonal Boron Nitride. J. Phys. Chem. C 2018, 122, 25524–25529.

83. Li, S.L.; Komatsu, K.; Nakaharai, S.; Lin, Y.F.; Yamamoto, M.; Duan, X.; Tsukagoshi, K. Thickness Scaling Effect on Interfacial Barrier and Electrical Contact to Two-Dimensional MoS₂ Layers. *ACS Nano* **2014**, *8*, 12836–12842.

84. Voiry, D.; Mohite, A.; Chhowalla, M. Phase Engineering of Transition Metal Dichalcogenides. *Chem. Soc. Rev.* 2015, 44, 2702–2712.

85. Kappera, R.; Voiry, D.; Yalcin, S.E.; Branch, B.; Gupta, G.; Mohite, A.D.; Chhowalla, M. Phase-Engineered Low-Resistance Contacts for Ultrathin MoS₂ Transistors. *Nat. Mater.* **2014**, *13*, 1128–1134.

86. Sun, L.; Yan, X.; Zheng, J.; Yu, H.; Lu, Z.; Gao, S.P.; Liu, L.; Pan, X.; Wang, D.; Wang, Z.; et al. Layer-Dependent Chemically Induced Phase Transition of Two-Dimensional MoS₂. *Nano Lett.* **2018**, *18*, 3435–3440.

87. Duerloo, K.A.N.; Li, Y.; Reed, E.J. Structural Phase Transitions in Two-Dimensional Mo- and W-Dichalcogenide Monolayers. *Nat. Commun.* **2014**, *5*, 4214.

88. Qi, D.; Wang, Q.; Han, C.; Jiang, J.; Zheng, Y.; Chen, W.; Zhang, W.; Wee, A.T.S. Reducing the Schottky Barrier between Few Layer MoTe² and Gold. 2D Mater. **2017**, *4*, 045016.

89. Cho, S.; Kim, S.; Kim, J.H.; Zhao, J.; Seok, J.; Keum, D.H.; Baik, J.; Choe, D.H.; Chang, K.J.; Suenaga, K.; et al. Phase Patterning for Ohmic Homojunction Contact in MoTe₂. *Science* **2015**, *349*, 625–628.

90. Addou, R.; McDonnell, S.; Barrera, D.; Guo, Z.; Azcatl, A.; Wang, J.; Zhu, H.; Hinkle, C.L.; Quevedo-Lopez, M.; Alshareef, H.N.; et al. Impurities and Electronic Property Variations of Natural MoS₂ Crystal Surfaces. ACS Nano **2015**, *9*, 9124–9133.

91. Guo, Y.; Liu, D.; Robertson, J. 3D Behavior of Schottky Barriers of 2D Transition-Metal Dichalcogenides. ACS Appl. Mater. Interfaces 2015, 7, 25709–25715.

92. Kerelsky, A.; Nipane, A.; Edelberg, D.; Wang, D.; Zhou, X.; Motmaendadgar, A.; Gao, H.; Xie, S.; Kang, K.; Park, J.; et al. Absence of a Band Gap at the Interface of a Metal and Highly Doped Monolayer MoS₂. *Nano Lett.* **2017**, *17*, 5962–5968.

93. Chee, S.S.; Seo, D.; Kim, H.; Jang, H.; Lee, S.; Moon, S.P.; Lee, K.H.; Kim, S.W.; Choi, H.; Ham, M.H. Lowering the Schottky Barrier Height by Graphene/Ag Electrodes for High-Mobility MoS₂ Field-Effect Transistors. *Adv. Mater.* **2019**, *31*, e1804422.

94. Song, S.; Sim, Y.; Kim, S.Y.; Kim, J.H.; Oh, I.; Na, W.; Lee, D.H.; Wang, J.; Yan, S.; Liu, Y.; et al. Wafer-Scale Production of Patterned Transition Metal Ditelluride Layers for Two-Dimensional Metal–Semiconductor Contacts at the Schottky–Mott Limit. *Nat. Electron.* **2020**, *3*, 207–215.

95. Lee, C.S.; Oh, S.J.; Heo, H.; Seo, S.Y.; Kim, J.Y.; Kim, H.; Kim, D.; Ngome, O.F.; Shin, H.; Sung, J.H. et al. Epitaxial Van Der Waals Contacts between Transition-Metal Dichalcogenide Monolayer Polymorphs. *Nano Lett.* **2019**, *19*, 1814–1820.

96. Shin, H.G.; Yoon, H.S.; Kim, J.S.; Kim, M.; Lim, J.Y.; Yu, S.; Park, J.H.; Yi, Y.; Kim, T.; Jun, S.C.; et al. Vertical and In-Plane Current Devices using NbS₂/N-MoS₂ Van Der Waals Schottky Junction and Graphene Contact. *Nano Lett.* **2018**, *18*, 1937–1945.

97. Jang, J.; Kim, Y.; Chee, S.S.; Kim, H.; Whang, D.; Kim, G.H.; Yun, S.J. Clean Interface Contact using a ZnO Interlayer for Low-Contact-Resistance MoS₂ Transistors. *ACS Appl. Mater. Interfaces* **2020**, *12*, 5031–5039.

98. Chen, J.R.; Odenthal, P.M.; Swartz, A.G.; Floyd, G.C.; Wen, H.; Luo, K.Y.; Kawakami, R.K. Control of Schottky Barriers in Single Layer Mos² Transistors with Ferromagnetic Contacts. *Nano Lett.* **2013**, *13*, 3106–3110.

99. Lee, S.; Tang, A.; Aloni, S.; Wong, H.S. Statistical Study on the Schottky Barrier Reduction of Tunneling Contacts to CVD Synthesized Mos2. *Nano Lett.* **2016**, *16*, 276–281.

100. Wang, J.; Yao, Q.; Huang, C.W.; Zou, X.; Liao, L.; Chen, S.; Fan, Z.; Zhang, K.; Wu, W.; Xiao, X. et al. High Mobility MoS² Transistor with Low Schottky Barrier Contact by using Atomic Thick H-BN As a Tunneling Layer. *Adv. Mater.* **2016**, *28*, 8302–8308.

101. Mleczko, M.J.; Yu, A.C.; Smyth, C.M.; Chen, V.; Shin, Y.C.; Chatterjee, S.; Tsai, Y.C.; Nishi, Y.; Wallace, R.M.; Pop, E. Contact Engineering High-Performance n-type MoTe₂ Transistors. *Nano Lett.* **2019**, *19*, 6352–6362.

102. Cho, K.; Pak, J.; Kim, J.K.; Kang, K.; Kim, T.Y.; Shin, J.; Choi, B.Y.; Chung, S.; Lee, T. Contact-Engineered Electrical Properties of MoS₂ Field-Effect Transistors via Selectively Deposited Thiol-molecules. *Adv. Mater.* **2018**, *30*, 1705540.

103. Wirth-Lima, A.J.; Alves-Sousa, P.P.; Bezerra-Fraga, W. Graphene/Silicon and 2D-MoS₂/Silicon Solar Cells: A Review. *Appl. Phys.* A 2019, 125, 241.

104. Sinha, D.; Lee, J.U. Ideal Graphene/Silicon Schottky Junction Diodes. Nano Lett. 2014, 14, 4660–4664.

105. Di Bartolomeo, A. Graphene Schottky Diodes: An Experimental Review of the Rectifying Graphene/Semiconductor Heterojunction. *Phys. Rep.* **2016**, *606*, 1–58. 106. Wong, H.; Gritsenko, V.A. Defects in Silicon Oxynitride Gate Dielectric Films. *Microelectron. Reliab.* **2002**, *42*, 597–605.

107. Lenahan, P.; Conley, J., Jr. What Can Electron Paramagnetic Resonance Tell Us about the Si/SiO₂ System? *J. Vac. Sci. Technol. B* **1998**, *16*, 2134–2153.

108. Wong, H.; Cheng, Y.C. Generation of Interface States at the Silicon/Oxide Interface due to Hot-Electron Injection. J. Appl. Phys. **1993**, 74, 7364–7368.

109. Glemža, J.; Palenskis, V.; Gudaitis, R.; Jankauskas, Š.; Guobienė, A.; Vasiliauskas, A.; Meškinis, Š.; Pralgauskaitė, S.; Matukas, J. Low-frequency Noise of Directly Synthesized Graphene/Si(100) Junction. *Diam. Relat. Mater.* **2022**, *127*, 109207.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.